
BRAHMS Cu+Cu Status
• ~1 nb-1 recorded since Jan. 16th

33M MRS 74M FFS Triggers collected
• 71 hours data taking for the last week 

(Wed.-Tues.):~42%
• “soft physic” (low-pT) program done
• Currently running high-pT program 

- y~1,2,3,(0)
- Goal: 4 nb-1 delivered 

• Complete installation of Shower Max Detector 
today.
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